Author's response to reviews

Title: New expectations in the prevention of pneumococcal diseases

Authors:

Catherine Weil-Olivier (cweilolivier@gmail.com)
Mark van der Linden (mlinden@ukaachen.de)
Iris de Schutter (Iris.DeSchutter@uzbrussel.be)
Ron Dagan (rdaqan@bgu.ac.il)
Lorenzo Mantovani (mantovani@unina.it)

Version: 2  Date: 21 February 2012

Author's response to reviews:

As requested by Dr. Diana Marshall, PhD, Deputy Managing Editor, we have included a reference for the source of the data in Figure 1 of our manuscript. The revised manuscript has been uploaded. The manuscript text (page 6, 3rd paragraph) and the Figure 1 legend have been revised to indicate the new reference. Thank you for your consideration of our manuscript.